Author's response to reviews

Title: Developing a scale to measure family dynamics related to long-term care, and testing that scale in a multicenter cross-sectional study

Authors:

Tesshu Kusaba (tesshu@hcfm.jp)
Kotaro Sato (kotaros@hcfm.jp)
Yoshinori Matsui (hcfdmrdr@gmail.com)
Satoshi Matsuda (satoshi.matsuda619@gmail.com)
Takashi Ando (takashi.ando@hcfm.jp)
Ken Sakushima (sakusima@med.hokudai.ac.jp)
Takafumi Wakita (wakita@kansai-u.ac.jp)
Shingo Fukuma (shingo.fukuma@gmail.com)
Shunichi Fukuhara (fukuhara.shunichi.6m@kyoto-u.ac.jp)

Version: 6  Date: 12 May 2014

Author's response to reviews:

We have revised the author's contribution according to your suggestion.